Author's response to reviews

Title: Breast Cancer Screening Disparities among Urban immigrants: A Population-Based Study in Ontario, Canada

Authors:

Mandana Vahabi (mvahabi@ryerson.ca)
Aisha K Lofters (aisha.lofters@utoronto.ca)
Matthew Kumar (mattkumar@gmail.com)
Richard H Glazier (rick.glazier@ices.on.ca)

Version: 6 Date: 3 July 2015

Author's response to reviews: see over
Thank you for your comments of July 3, 2015 regarding our manuscript “Breast Cancer screening Disparities among Urban Immigrants: A Population-Based Study in Ontario” (MS:160291477142983). We have addressed the one issue which was raised by Reviewer 3 and highlighted it in yellow: suggested to add the citation to the statement captured in Lines 139-140. This has been added and highlighted in yellow.

**Referee 3**

Title: Breast Cancer Screening Disparities among Urban immigrants: A Population-Based Study in Ontario, Canada

Date: 30 June 2015

Reviewer: Hung-Wen Yeh

Reviewer's report:

**Minor Essential Revisions**

Line 139-140. Please add the citation to “These variables were selected a priori based on literature review.”  **Response: The citation has been added. Please see Line 140 highlighted in Yellow.**